Author’s response to reviews

Title: Towards a standardised approach for evaluating guidelines on Palliative Sedation: Study Protocol

Authors:

Ebun Ebun Abarshi (e.abarshi@lancaster.ac.uk)
Judith Rietjens (j.rietjens@erasmusmc.nl)
Augusto Caraceni (Augusto.Caraceni@istitutotumori.mi.it)
Sheila Payne (s.a.payne@lancaster.ac.uk)
Luc Deliens (luc.deliens@vub.ac.be)
Lieve Van Den Block (lvdblock@vub.ac.be)

Version: 2 Date: 8 December 2013

Author’s response to reviews:

This revision explains the following:

1. This study is unfunded and has therefore not been peer-reviewed.

2. Ethical approval was not required for conducting it.

3. Data collection is on-going and we have neither published nor submitted any findings to any other journal.